CSE 521: Design and Analysis of Algorithms I

Win ter 1999

Administrativ e Information

Instructor:

Martin T ompa

tompa@cs.w ash ington.edu

Sieg 426B

543-9263

T eac hing Assistan t:

Saurabh Sinha

saurabh@cs.w ashington.e du

O�ce hours: Will b e announced shortly .

References (only the �rst is required):

1. Thomas H. Cormen, Charles E. Leisers on, and Ronald L. Riv est

Intr o duction to A lgorithms

MIT Press and McGra w-Hill, 1990

(course text)

2. Alfred V. Aho, John E. Hop croft, and Je�rey D. Ullman

The Design and A nalysis of Computer A lgorithms

Addison-W esley

, 1974

3. Ellis Horo witz and Sarta j Sahni

F undamentals of Computer A lgorithms

Computer Science Press, 1978

4. Donald E. Kn uth

The A rt of Computer Pr o gr amming

Addison-W esley

, 1968, 1969, 1973, : : :

5. John D. Lipson

Elements of A lgebr a and A lgebr aic Computing

Addison-W esley

, 1981

6. Rob ert E. T arjan

Data Structur es and Network A lgorithms

So ciet y for Industrial and Applied Mathematics, 1983.

Reading assignm en t: In Cormen, Leisers on, and Riv est, read Chapters

1 { 5. Y ou ma y skip Section 4.4 if y ou lik e, but the rest is in v aluable bac k-

ground.

Grading: The course grade will b e based on assignmen ts, a �nal exam,

and class participation. The �nal exam is sc hedu led for 10:30 a.m. { 12:20

p.m. on Monda y , Marc h 15.

Mailing List: The class mailing list is cse521@cs. wash ing ton. edu. W e

will use the list for announcemen ts of general in terest to the class. Y ou

should also feel free to use it to ask questions, p ost information, or initiate

discuss ions of general in terest to the class. Instructions for subscribi ng are

on the course w eb.

Course W eb: All handouts and a log of all mail sen t to the class

mailing list will also b e a v ailable on the course w eb at the URL

http://w ww.c s.wa shin gto n.ed u/ed ucat ion/ cou rses /521 /Cur ren tQtr /.

If y ou w ould lik e to request academic accommo dations due to a disabilit y ,

please con tact Disabled Studen t Services , 448 Sc hmitz, 543{8924 (V/TDD).

If y ou already ha v e a letter from Disabled Studen t Services indicating y ou

ha v e a disabilit y that requires academic accommo dations, please let the in-

structor kno w so w e can discuss the appropriate accommo dations.